Author's response to reviews

Title: A randomised controlled trial to assess the effectiveness of a single session of nurse administered massage for short term relief of chronic non-malignant pain

Authors:

Kate Seers (kate.seers@warwick.ac.uk)
Nicola Crichton (crichtn@lsbu.ac.uk)
June Martin (june.martin@orh.nhs.uk)
Katrina Coulson (katrina.coulson@derbyhospitals.nhs.uk)
Dawn Carroll (dawncarroll@btinternet.com)

Version: 4 Date: 28 April 2008

Author's response to reviews:

Thank-you.

Dan Cherkin comments:
1) I have made the revisions suggested on track changes as they are so few and difficult to highlight any other way. I have checked the aims in the methods section (p5) and it clearly states short term relief and a single session, so have not altered this.
2) I have deleted the section on the literature search and 1st 3 sentences of the next paragraph as requested.

Reviewer 2 - Andrea Furlan. Is happy with the answers we have given.